Author's response to reviews

Title: Rhodiola crenulata Extract for Prevention of Acute Mountain Sickness: A Randomized, Double-Blind, Placebo-Controlled, Crossover Trial

Authors:

Te-Fa Chiu (tefachiu@gmail.com)
Lisa Li-Chuan Chen (lcchen1216@gmail.com)
Deng-Huang Su (dhs@ntu.edu.tw)
Hsiang-Yun Lo (seanlo@adm.cgmh.org.tw)
Chung-Hsien Chen (chenchunghsien@gmail.com)
Shih-Hao Wang (er.mountain.tw@gmail.com)
Wei-Lung Chen (er.mountain.gogo@gmail.com)

Version: 2 Date: 9 August 2013

Author's response to reviews:

We would like to express our deepest gratitude to you for the constructive comments and suggestions to our manuscript. We have completed the revision of the manuscript based on all the reviewers’ general and specific comments. As indicated below, we have made necessary changes according to the comments and suggestions. We used blue color text for all changes in this revision. We look forward to hearing from you soon.